Author's response to reviews

Title: What factors are associated with recent intimate partner violence? Findings from the WHO multi-country study on women's health and domestic violence

Authors:

Tanya Abramsky (tanya.abramsky@lshtm.ac.uk)
Charlotte H Watts (charlotte.watts@lshtm.ac.uk)
Claudia Garcia-Moreno (garciamorenoc@who.int)
Karen Devries (karen.devries@lshtm.ac.uk)
Ligia Kiss (ligia.kiss@lshtm.ac.uk)
Mary Ellsberg (mellsberg@icrw.org)
Henriette AFM Jansen (henriette.jansen@gmail.com)
Lori Heise (lori.heise@lshtm.ac.uk)

Version: 5 Date: 14 February 2011

Author's response to reviews: see over
Dear Editor,

I have made the requested changes to the formatting of our paper: “What factors are associated with recent intimate partner violence? Findings from the WHO multi-country study on women’s health and domestic violence”. Please find these changes detailed below:

1) The authors in the manuscript now match exactly those in the submission system.

2) The abstract is present in the manuscript and matches that in the submission system.

3) The footnote now appears as an endnote in an ‘Endnotes’ section that appears just before the references section.

4) Tables 1a and 1b have been merged into one table (Table 1). The same has been done for Tables 2a and 2b (Table 2) and Tables 3a and 3b (Table 3). They are referred to in the text accordingly.

5) The Figures have been renumbered so they are now cited in order in the text.

I hope that the paper is now ready for publication, but please get in touch if anything is not clear or if further changes are needed.

Yours sincerely,

Tanya Abramsky
Research Fellow
Centre for Gender, Violence and Health
London School of Hygiene and Tropical Medicine